TRANSFORMATIONAL LEADERSHIP:

AN INVESTIGATION INTO THE EXISTENCE OF

TRANSFORMATIONAL LEADERSHIP AT A DURBAN SECONDARY SCHOOL

BY

ALISTAIR R. KOK

SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS

FOR THE DEGREE OF MASTER OF

EDUCATION (EDUCATIONAL MANAGEMENT) IN THE

SCHOOL OF EDUCATION

UNIVERSITY OF KWAZULU NATAL

DURBAN

DECEMBER 2004

ABSTRACT

Coinciding with a rebirth in South Africa, new education legislation has given considerable more power and responsibility for making decisions. While assuming a degree of autonomy, schools are also expected to resource themselves with the minimum support from their respective provincial departments. The extent to which schools will be able to make the transition depends a great deal upon the nature and quality of their internal management. (Department of Education 1996). The Task Team on Education Management Development (1996) says that a move to self-management does not mean that you are guaranteed positive change. It is the nature and quality of internal management that will influence real transformation. So, a self-managing school is accompanied by an internal devolution of power and by a leadership style that is transformational.

It is here that an attempt is made to look at the extent to which transformational leadership is practiced in a Durban Secondary School. Seeing that transformational leadership is linked with school effectiveness, (Coleman, 1994) it is the aim of this research project to evaluate the principal's leadership style. Does the way the school is managed contain any evidence of transformational leadership? Is the principal's perception of his leadership style parallel with the staff's perception of how he leads the school?

The findings of this research have shown that there is no evidence of transformational leadership at this school. The recommendations have highlighted ways in which the principal can begin to transform the school so that they too may improve to such an extent that the school is seen as being effective.

DECLARATION

declare that this work is my own and where other sources nave been consulted, these have been acknowledged through references.

A.R.KØK

DECEMBER 2004

SUPERVISOR'S STATEMENT

this mini dissertation is submitted with / without my approval.

PROFESSOR M. THURLOW

DEDICATION

This project is dedicated to my wife Beverley, my children Leah and Joshua and my family and close friends who have supported me and given me strength during these years of study.

ACKNOWLEDGEMENTS

Thank you to Almighty God, without whom nothing is possible. A special thanks to the principal and staff of this school for their cooperation in this research project; without you this project would not have succeeded. I am eternally grateful to my Supervisor, Professor M. Thurlow who has guided and supported me throughout this research project.

TABLE OF CONTENTS

	<u>PAGES</u>
Abstract	i
Declaration	ii
Supervisors Statement	iii
Dedication	iv
Acknowledgements	V
able of Contents	vi
ist of Tables and Figures	Vìíi
CHAPTER 1:INTRODUCATION AND OVERVIEW	
.1.Introduction	1
.2. Context of the Study	2
.3. The Focus of the Study	2
.4. Research Methods	2 2 3 4
.5. Limitations of the Study	4
.6. The Structure of the Study	4
.7. Conclusion	5
CHAPTER 2: LITERATURE REVIEW	
2.1. Introduction	6
2.2. Leadership: Is there a Definition?	6 7
2.3. Theories of Leadership	
2.4. Leadership and Management	9
2.5. Characteristics of a Good Leader	12
2.6. Transactional and Transformational Leadership	16
2.7. School Effectiveness and Management	19
2.8. The Context of Management in South African Schools	21
2.9. Conclusion	23
CHAPTER 3: RESEARCH METHODS	
3.1. Introduction	25
3.2. The Aims of the Research	25
3.3. Research Questions	25
3.4. Research Methods	26
3.5. Research Population	26
3.6. The Research Instruments	
3.7. Data Analysis	30
3.8. Conclusion	30

CHAPTER 4: THE FINDINGS OF THE RESEARCH	
f.1. Introduction	32
1.2. The Questionnaire	32
4.3. Summary	38
1.4. The Interview	39
1.5. Overview of the Findings	43
1.6. Conclusion	44
CHAPTER 5: THE FINDINGS OF THE RESEARCH AND RECOMMENDATIONS	
5.1. Introduction	45
5.2. Summary of the Findings	45
5.3. Recommendations	48
5.4. Conclusion	49
References	50
ADDENDUM 1: Questionnaire	54
ADDENDUM 2: Interview Schedule (Principal)	58
ADDENDUM 3: Interview Schedule (Deputy-Principal, lead of Department & Educator)	59

LIST OF TABLES AND FIGURES

FIGURE	TITLE	PAGE
2.1.	The Leader	19

CHAPTER 1: INTRODUCTION AND OVERVIEW

.1. Introduction

rior to the birth of the 'New South Africa' in 1994, the South African education system was divided into a number of racially defined education departments. White schools fell unde the governance of the House of Assembly and they enjoyed the lion's share of the allocated education resources while the Coloured, Indian and Black schools were ruled be the House of Representatives, House of Delegates and the Department of Education and raining respectively. The latter majority enjoyed very little in the way of resources and under the apartheid government, a bureaucratic and authoritarian management style vas the order of the day.

Coinciding with a rebirth in South Africa, new education legislation has given considerably more power and responsibility for making decisions. While assuming a degree of autonomy, schools are also expected to resource themselves with the minimum support rom their respective provincial departments. The extent to which schools will be able to make the transition depends a great deal upon the nature and quality of their internal management. (Department of Education, 1996). The move to decentralize education systems is not a new one because countries like England, Australia and the United States of America have embraced the notion of self-managing schools. Bush (1995: pg4) sees hese autonomous schools as "...potentially more efficient and effective but much depends on the nature and quality of internal management if the ... benefits of autonomy are to be realized." Echoing this view is the Task Team on Education Management Development (1996) saying that a move to self-management does not mean that you are guaranteed positive change. It is the nature and quality of internal management

affluence real transformation. So, a self-managing school is accompanied by an internal devolution of power and by transformational leadership.

t is against this backdrop that the student attempts to look at the extent to which ransformational leadership is practiced in a Durban Secondary School. Seeing that ransformational leadership is linked with school effectiveness, (Coleman, 1994) it is the ain of this research project to evaluate the principal's leadership style. Does the way the chool is managed contain any evidence of transformational leadership? Is the principal's perception of his leadership style parallel with the staff's perception of how he leads the chool?

.2. Context of the Study

his study focuses on a secondary school in central Durban, KwaZulu Natal having a staff complement made up of a principal, 2 deputies, 5 heads of department and 25 level one educators. The staff members are all qualified educators, with qualifications ranging from three -year education diploma to post-graduate degrees. More than seventy percent of the staff has been teaching for more than ten years.

he school chosen for the study is one in which the researcher is a member of staff. This esearch hopes to highlight the quality of leadership at this school, the findings of which are intended to help improve the quality of the leadership at this school. The study's pecific focus is on the principal as a leader, asking to what extent he can be called a ransformational leader.

.3. The Focus of the Study

the study attempts to highlight what is understood by the term transformational leadership and its significance in the context of South African schools. Then, it will attempt to

tetermine the principal's perception of himself as a leader, the educators' perception of im as a leader and a comparison of the two perceptions. The research looks at the ollowing critical questions:

- .What is understood by the term Transformational Leadership?
- !.What is its significance in the context of South African schools?
- i. What is the principal's perception of his leadership style and to what extent does this style and leadership?
- . How do the educators perceive the principal's leadership style? Is there congruence between the two perceptions?
- i. Based upon these perceptions, to what extent is the leadership at this school ransformational?
- . What are the implications of the findings for possible leadership development in the chool?

.4. Research Methods

n conducting this research, the methods of collecting data would be both qualitative and quantitative, namely a semi-structured interview schedule and a questionnaire.

Arrangements were made for the principal, a deputy, one head of department and a positive one educator to be interviewed. The remainder of the staff were asked to complete questionnaire supplying information regarding their perceptions regarding the principal's eadership style. In an attempt to address the key questions, the following methods will be used:

Question 1 and 2 will be answered by reviewing the relevant international and local literature regarding transformational leadership.

Question 3 requires a semi-structured interview with the school principal while question will be answered by means of an interview with one deputy principal, one HOD and one level one educator. The remainder of the staff will require a questionnaire to be completed.

he answer to questions 5 and 6 is addressed by referring to questions 3 and 4 after ne collected data is analysed and interpreted.

.5. Limitations of the Study

he findings of this study are limited to the school in which the study is conducted. In no vay can these be generalised to any other cases. The concept of leadership is a broad one and has been researched widely. There are therefore many varying opinions regarding good leadership. This study's opinions are derived from a number of sources, but these opinions are just that.

.6. The Structure of the Study

his project has been divided into five chapters.

Chapter 1 is an introductory chapter, providing a brief overview of what the study is all about, namely the concept of leadership, what research methods were used and the imitations of the study.

Chapter 2 examines the literature connected to leadership, referring to transformational eadership specifically. This chapter looks at definitions of leadership, theories of leadership.

ne difference between leadership and management, characteristics of a good leader, ransactional and transformational leadership, school effectiveness and management and ne context of management in South African schools.

Chapter 3 deals with the research methods employed to conduct the research. The aims of the research are explained before examining the actual methods used, the research questions and the research instruments.

Chapter 4 presents the findings of the research in which respondents were interviewed or equired to fill out a questionnaire.

Chapter 5 summarises the findings of the research and makes recommendations on how 1 mprove leadership at this school.

.7. Conclusion

the research project is aimed to look at leadership in a Durban-central secondary school. The next chapter examines the literature on leadership, focusing specifically on the terature directly connected to this study.

CHAPTER 2: LITERATURE REVIEW

.1. Introduction

ne concept of leadership is quite complex with many researchers giving varying lefinitions of what it is. There is definitely no one cut and dried definition of leadership, eing viewed differently in different cultures. In reviewing the literature on leadership, we ook at the various definitions and theories of leadership before examining the link etween leadership and management. Effective leaders would have certain qualities so then tion would be focussed upon the role of transformational leaders as opposed to ansactional leaders in South African schools post-1994.

.2. Leadership: Is there a definition?

seems that everyone knows what leadership is, but when asked to define it, then each efinition differs. Stogdill (1974) points out that there are as many different views of adership as there are people trying to define it. There are, however, many similarities in the definitions that follow.

Leadership is the exercise of influence in a group context. ...(They are) the people having ne greatest impact on the group or organisation." (Johnson, 2001: 6)

The common characteristic of ... leaders is their ability to inspire and stimulate others to chieve worthwhile goals. ...(L)eadership is the ability to inspire confidence and support mong the people who are needed to achieve organisational goals." (Dubrin, 2001:3) owalski and Reitzug (1993) define leadership as "... a process that results in the etermination of organisational objectives and strategies, entails building consensus to

neet those objectives and involves influencing others to work towards those objectives. Squelch & Lemmer, 1994: 2)

nere are numerous ways of conceptualising what leadership is, but we can isolate a umber of components that are central to the phenomenon of leadership. Northouse 2001) says that leadership is a process. It is not a linear, one-way event, but an interactive ne between the leader and his followers. Secondly, leadership involves *influence*; without thich the phenomenon of leadership would not exist. Thirdly, leadership occurs in *groups*, ne context in which leadership takes place. Leaders influence these groups to achieve or rork towards a *common goal*.

.3. Theories of Leadership

ait Approach to Leadership

ne trait perspective on leadership suggests that certain individuals have certain innate haracteristics that make them leaders. These characteristics make them different from on-leaders. Some of the differentiating characteristics include unique physical factors, ersonality features, and ability characteristics. It is believed that the special leadership roperties reside in selected people and leadership is restricted to only these people.

rocess Leadership

nis viewpoint suggests that everyone possesses leadership qualities. As a process, adership can be observed in the behaviour of leaders and it is something that can be earned.

ssigned Leadership

nis approach suggests that some people are leaders because of their formal position that new occupy within an organisation, for example, team leaders, heads of department, tc.

mergent Leadership

Then others see an individual as being the most influential member of a group or rganisation, regardless of their title, then that person is exhibiting emergent leadership.

The organisation of department.

lyle Approach

ne style approach focuses on what leaders do and how they act. Here leadership is iewed as being composed of two general kinds of behaviours namely, task behaviours nd relationship behaviours. Task behaviours facilitate the accomplishment of goals, elping group members to achieve their objectives. Relationship behaviours help ubordinates feel comfortable with themselves, with each other and with the situation that ney find themselves in. This approach aims to explain how leaders combine these two roes of behaviour to influence subordinates to reach a particular goal.

tuational Approach

tuational leadership stresses that leadership is composed of both a directive and a upportive dimension, each to be applied appropriately in a given situation. The essence

f situational leadership demands a leader's matching his or her style to the competence nd commitment of the subordinates. Effective leaders are seen to be those who can accognise what employees need and then adapt their own style to meet those needs.

ansactional Leadership

efers to the bulk of leadership models that focus on the exchanges that occur between aders and their followers. For example, teachers are being transactional when they give udents a grade for work completed. Emphasis here is on the exchange dimension.

ansformational Leadership

hereby an individual engages with others and creates a connection that raises the level f motivation and morality in both the leader and follower. The transformational leader is ttentive to the needs and motives of his or her followers, trying to help them realise their all potential.

.4. Leadership and Management

Leadership and management are not synonymous terms. One can be a leader without being a manager. One can, for example, fulfil many of the symbolic, inspirational, educational and normative functions of a leader and thus represent what an organisation stands for without carrying any of the formal burdens of management. Conversely, one can manage without leading. An individual can monitor and control organisational activities, make decisions, and allocate

resources without fulfilling the symbolic, normative, inspirational, or educational functions of leadership (Schon 1983, p36).

an attempt to understand leadership, it is important to see how it differs from nanagement: planning, organising, directing and controlling. It seems that leading is a najor part of a manager's job, but he also has to plan, organise and control. Leadership rould deal with the interpersonal aspects of a manager's job whereas planning, rganising and controlling deal with the administrative aspects. Leadership deals with hange, inspiration, motivation and influence. Sterling et al (2000), points out the ifferences between leadership and management:

- A leader guides while the manager would coordinate.
- Leaders motivate their subjects and managers organise.
- The leader is the initiator while the manager maintains.
- Leaders anticipate and managers stabilise.
- A leader would build a vision for the school and the manager helps the school to realise that vision.
- Leaders create and managers structure.
- Leaders are concerned with moving forward and managers with the establishment of parameters.
- Leaders inspire. Managers handle.
- A leader breaks boundaries while managers set them.

thich of the two concepts are more important in the school context? Sterling et al (2000) ays that the two work together, being two sides of the same coin. A teacher in a adership position cannot be effective as a leader if he/she is an incompetent manager.

milarly, the managerial work of someone who coordinates day-to-day functioning of the chool is undermined if there is no holistic view of the long-term development plan of the chool. So, a leader needs both leadership and managerial skills to be effective as a eader in the school context.

is clear that the functions of leadership and management cannot be separated ecause there are certain functions in the school that are both leadership and nanagement functions.

rerling et al (2000) assigns functions like strategic planning, building of the school's vision, aison with communities and parents and development of the staff can be identified as the functions of a leader, while the managerial functions would include staff meetings, but the utility rosters and administering textbooks. If these two entities were separate domains, then the intersection of the two would include decision-making, delegating, conflict resolution and staff appraisal. So, to be an effective leader, effective management skills are a rerequisite if you really want to make a difference at your school.

em to tactical and strategic leadership. Tactical leadership being a small-scale analysis at leads to administrative action while the strategic identifies with beliefs and ommitment. Leaders are thus charged with a number of responsibilities that have great applications for the improvement of the institution, the most important one being eveloping and communicating a vision for the school.

wens (1998) says that leadership and management are not only different, but are utually exclusive.

This view correctly derives from the fact that one manages things, not people, and one leads people, not things. We manage finances, inventories, and programs, for

example but we lead people.

Owens, 1998: pg 21&

ven though there is a dire need for leadership at all levels of educational institutions, aution should be taken about "...substituting management-bashing for leadership."

Dwens, 1998: pg 217). Educational leaders have got to be able to manage. They have allocate resources, deal with budgets as well as organise the people in the institution to the necessary work so that they move towards attaining its vision. So, if a leader is a filled manager, then he/she is able to deal with mundane, everyday workings of the reganisation while striving to reach their vision.

5. The Characteristics of a Good Leader

5.1. Planning and Vision

ntil recently, school leaders have not seen planning ahead to be a priority. Taking a longerm view of the school was unheard of in school managerial circles. Instead, school lanagers were preoccupied with operational issues, solving problems as they presented lemselves.

is lack of attention to planning is now rapidly disappearing because of the importance is trategic thinking and forward planning for school leaders becomes inevitable as they assume responsibility for their own budgets and school resources. Schools are now aliged to present a School Management Plan, setting out priorities for development and aw they go about implementing them as well as anticipating future resources. However, reating a plan is not the same as creating a vision (Jenkins, 1991). Planning has to do with the problems of today and tomorrow, prioritising and allocating resources on a short-term

ision is an interpretation of the direction in which an organisation is going, something ecessary to give people a belief in the organisation's future. It is not an imaginary picture, ut "...a view of a realistic, credible, attractive future for the organisation" (Bennis and anus, 1985:89). Jenkins (1991) says that a vision does the following for a school:

- It inspires. A vision transforms the ordinary school life into something special.
- It binds the school together, incorporating shared beliefs of what the school is about.
- Its clarity and power strengthens the bonds the school has with the broader school community.
- It gives staff a view of excellence of what the school is capable of achieving.
- It lights the way ahead, giving direction during trying times.
- A vision places school staff at the centre of things. It is about what people can achieve and the extent to which people are valued in an organisation.

nere are a few important points to remember as a principal when attempting to implement a successful vision. The first is that the vision has to be owned by all the akeholders. A vision cannot be imposed. Bennis and Nanus (1985) warn that a vision cannot be created successfully if it grows out of the needs of the entire organisation and is when when when the important stakeholders. Secondly, constantly articulate and clarify the sion, modifying it if need be. Thirdly, remember that a vision does not just happen vernight. Spend time on the vision remembering that it is a continuous process that is ever-ending. Every opportunity that you get to refer to the vision, hammer the message

ome. Lastly, make sure that you enact the vision, embedding it in the culture of the chool. In all the school structures, the school's vision would be omnipresent. When making ay-to-day decisions, bear in mind the vision of the school, bearing in mind that you are onstantly under scrutiny. If you as the leader don't enact the vision, then other school akeholders would become disillusioned.

Indeed, one of the pivotal activities of leaders is to engage constantly in the dynamic process of stating a vision of things to come; then revising in light of emerging events, ideas and beliefs; and restating the vision of "where we are and where we are going" that coheres the members of the organisation in mutual purpose and resolve.

Owens R. G., 1998: pg212

eare et al (1993, pg 157) emphasises the link between leadership and vision when he says nat:

The term 'vision' ... is not describing a new phenomenon in leadership. It is simply attaching a label to the sort of dream or constellation of goals or scenarios that form in the mind of everyone from time to time. What we now know is that these form readily in the minds of leaders who succeed in transforming their organisations.

.5.2. Empowering Staff

tany schools still have over-developed hierarchies and school leaders are still quite ominant figures. These hierarchies demotivate staff and stifle their creativity. A good chool leader would therefore move away from the token attempts at staff involvement to note radical forms of power sharing. A way to do this would be to create highly utonomous teams to whom power is granted on a self-managing basis. The leader will

ccept letting go of his/her power by playing the role of post-heroic leader. Teams rather ian dominant individuals will be celebrated as the 'heroes', being trained to perform at a gh level of efficiency. School leadership will then have to play the role of consultant, icilitator, counsellor and empowerer. The school leader would be charged with the task of ensuring that teams work effectively, while giving them the necessary back up and apport in terms of time and technical resources.

.5.3. Create fair and caring systems

put it simply, a good leader is consistently fair and takes care of his/her staff. To do this, e/she will need to have integrity and trust, a caring attitude to staff and reduce the micro folitics in the school. Staffs often feel uncared for and undervalued, so leaders must insure that all staff is treated fairly with proper care and consideration. A good leader tims to end interdepartmental rivalries, stop staff members blaming and criticising each other, remove the tendency to resort to manipulation and politics and the desire to overcontrol staff. Instead, work with people in such a way that nobody gets hurt and you build rust and support (Jenkins, 1991).

2.5.4. Offers strong instructional leadership

nstructional leadership is the one professional activity that marks out the distinctiveness of a school leader. The leader has a central role at school and that is to ensure that quality earning is offered to pupils. School leaders need to return urgently to strong instructional eadership as a core element in their activities. How is this done? The leader must be able o articulate a coherent learning philosophy while having a good understanding of the practical and theoretical issues that underpin curriculum, learning and assessment. He/

ne will be up to date with the latest thinking on curriculum having a thorough anderstanding of new learning developments. Lastly, it is his/her responsibility to create a isciplined climate in which effective learning can take place.

6. Transactional and Transformational Leadership

6.1. Transactional Leadership

ansactional leadership refers to the bulk of leadership models, which focus on the changes that occur between leaders and their followers. (Northhouse, 2001:32) ansactional leadership is about responsibility and loyalty and the consideration of others. enkins, 1991:18). Much of leaders' activities are transactional in that they involve accessary operational activities, getting things done and sorted out within the ganisation. A transactional leader would exchange things of value with subordinates to dvance their own as well as their subordinates' agenda (Kuhnert, 1994). These leaders the influential only because it is in their best interest of the subordinates to do what the ader wants.

is type of leadership is characterized by two factors. First there is an exchange process, here followers are rewarded for their efforts. Leaders would try to obtain agreement from llowers on what needs to be done and the payoff for the followers doing it. The second ctor according to Northouse (2001) is known as management by exception that involves prective criticism, negative feedback and negative reinforcement, for example, a ader watching followers closely for mistakes or rule violations and only then taking prective action. A transactional leader would only intervene after standards have not sen met or when problems have arisen.

6.2. Transformational Leadership

ansformational leaders engage with others and in so doing create a connection that tises the level of motivation and morality in both the leader and follower. Jenkins (1999) choes the definition given by Northhouse (2001), "Transformational Leadership...builds on eople's needs for meaning and for purpose in organisational life, and involves leaders and followers engaging in such a way as to lift one another to higher levels of motivation and morality" (pg18). It is Bass (1985) who further argues that transformational leadership notivates followers to do more than what is expected of them by:

- Raising the level of consciousness of followers concerning the importance and values of specified and idealized goals
- Getting the followers to put their own interests aside for the sake of the team or the organisation.
- Moving followers to address higher level needs.

hat is it that transformational leaders have? Northouse (2001) says that the ansformational leader has charisma or idealised influence acting as a strong role model r followers. Followers in turn identify with them and want very much to emulate them.

ass.(1990:22) defines charisma as that which, "...(P)rovides vision and sense of mission, stills pride, gains respect and trust." Northouse (2001) agrees that this charismatic leader ovides his/her followers with a vision and a sense of mission. Secondly, the ansformational leader communicates high expectations to followers, inspiring them rough motivation to become committed to and a part of the organisation's shared sion. This type of leadership stimulates followers to be creative and innovative, nallenging their own beliefs as well as those of the leader and the organisation. This

tellectual stimulation promotes followers thinking things out on their own, engaging in areful problem solving.

ne last transformational factor highlighted by Northouse (2001) is called individual onsideration. Here leaders provide a supportive climate where they listen carefully to the dividual needs of followers. Leaders are councillors and advisors, assisting individuals to ecome fully actualised.

is expected that transformational leaders produce greater effects than transactional radership. Yes, transactional leadership results in expected outcomes, but ansformational leadership results in performance that goes well beyond what is xpected. Transformational leadership moves followers to accomplish more than what is sually expected of them. Bennis and Nanus (1985) have virtually echoed views of the revious researchers. Transformational leaders had a clear vision of the future state of their rganisation; secondly they felt that transformational leaders were social architects for neir organisations, communicating a direction that transformed the organisations value and norms. Thirdly, transforming leaders created trust in their organisation by making their wn position clearly known and then standing by them. Trust in organisations gives the rganisation a sense of integrity. Lastly transforming leaders used creative deployment of elf through positive self-regards. They know the strengths and weaknesses of the workers mphasising the former rather than dwelling on the latter.

o if transformational leadership's main focus is on what leaders accomplish, hopefully najor positive change, Dabrin(2001) suggests seven ways in which transformation takes lace illustrated in the figure below(See Fig. 2.1)

THE LEADER

- 1. Raises people's awareness
- 2. Helps people look beyond self-interest
- 3. Helps people search for self-fulfilment
- 4. Helps people understand the need for change
- 5. Invests managers with a sense of urgency
- 6. Is committed to greatness
- 7. Adopts a long-range broad perspective

FIGURE 2.1

ansformational leadership is undoubtedly the newest and most encompassed approach leadership, looking at how some leaders are able to inspire followers to accomplish reat things. These leaders are recognized as being change agents, positive role models ho articulate a clear vision for the organisation. Transformational leadership emphasises in importance of followers in the leadership process, going beyond traditional ansactional models to include growth of followers and placing strong emphasis on lorals and values.

7. School Effectiveness and Management

7.1. Introduction

ne concept of school effectiveness is seen by many as being central to the progress of ducation in South African schools. The basic premise here is that there are many schools at are equipped with similar resources as other schools, produce better results and have different effect on their learners' development. Reynolds and Creemers (1990) say that

chools actually do make a difference. What is it that makes a school effective? Does the chool management and leadership contribute towards school effectiveness?

7.2. The Concept of School Effectiveness

efinitions of effectiveness differ greatly amongst educational researchers. However, mply put, effectiveness is the extent to which a school achieves its goals. No, the main oal of all schools is not only to achieve outstanding academic results. It includes learners' xperiences out of the classroom as well. Creemers and Reezigt (1997) say that it refers to Il theories and research studies that explain why student achievement between schools and classrooms differ.

Being effective is reflected in the institution nurturing a culture of teaching and learning. Research has shown that "...certain internal conditions are typical in schools that achieve higher levels of outcomes for their students." (Hargreaves et al., 1993;229).

nese internal conditions include the management of the most important resource in the stitution, people. Squelch and Lemmer (1994) see leadership being an important ontributor to the effectiveness of a school. School, staff and pupil performance is greatly feeted by the leadership role of the principal. Being the school principal does not mean at you are an effective leader. Rather, many principals acquire leadership skills as they along and even improve on them. How would they know that their school is an fective one?

.7.3. The Characteristics of an Effective School

school would be regarded as being effective if, according to Ubben & Hughes (1992), ney display the following:

mphasis on achievement. An effective school will achieve high academic results.

uilding a positive learning climate. Establishing a positive learning culture and nvironment is essential in an effective school. This depends largely on the attitudes of ducator, learners and parents towards the learning process and education in general.

nsure safety and order in the school. The school is kept free from disruption, chaos and anger.

lonitoring the progress of students continuously. If the progress of students is constantly nonitored, performance could be improved and areas that need attention could be lentified.

collegiality. It is the task of school leadership to develop positive staff attitudes and ollegiality. Motivate, support staff, involve them in decision-making, promote teamwork, is still trust and confidence and show them that their work is appreciated.

chools have good, strong, effective leaders. It may seem relatively simple to identify and it leadership tasks, but it is another for these to be fulfilled in practice, possible with ffective leadership.

8. The Context of Management in South African Schools

The legacy of apartheid in the field of education is well-known; it has left the country with an education system that is characterised by fragmentation, inequity in

provision, a crisis of legitimacy and in many schools, the demise of a culture of learning as well. (Report of the Task Team on Educational Management Development, 1996:10)

rior to the birth of the New South Africa in 1994, the school principal's role was different om the role that he/she now has to play. Where did this come from? arlier examples of schools in the United Kingdom, upon which the South African ducation system was based, saw school as social institutions that had somebody in harge. The head teacher was looked upon as being the leader of the institution, xercising some responsibility for setting and implementing standards. (Dunford et al, 2000) It that time, the early to middle 20th century, schools were, however, simple institutions whose primary task was to teach. During the years of apartheid people in leadership ositions were actually trained in rigid, bureaucratic management styles, their primary genda being to keep the school under control. Principals were primarily tasked with naintenance of the status quo.

Ifter 1994 and within the broad context of transformation, the National Department of ducation attempted to address the past imbalances through a series of policy initiatives, a line with the new Constitution. The Task Team on Education Management Development dentifies that:

The key challenge for education management relates to the inappropriate nature of many of the existing management systems, processes and structures. New education policy requires managers who are able to work in democratic and

1

participative ways to build relationships and ensure efficient and effective delivery.

(Department of Education, 1996b: 25)

ne school manager is now faced with far greater responsibility than ever before. Schools ow need effective and efficient school s management that are equipped with new and nproved skills, knowledge and attitudes to help them cope with a range of new emands and challenges. Managers now have to work more closely with parents and the ommunity at large, assume greater financial responsibility, cope with a multicultural chool population, and manage change and conflict while making do with fewer sources.

ne above challenges to school management is as a result of the South African Schools ct of 1996 ensuring that schools become self-managing institutions. "The extent to which nese schools are able to make the necessary changes depends largely on the nature nd quality of their internal management." (Department of Education, 1996: 28). It is the lace of transformational leaders to take up the challenge facing South African school rincipals. These leaders are visionaries, masterful communicators with the ability to inspire ust in their followers while making them feel capable. The terms leadership and nanagement, sometimes used interchangeably, are not synonymous. Schools in our ountry need principals who portray both qualities so that schools may realise their visions.

.9. Conclusion

y.c

that has clearly emerged from the leadership review is that South African schools need ansformational leaders to be effective schools. School principals can no longer be the

ureaucratic, controlling manager who is obsessed with hierarchies and maintaining the atus quo. School principals as transformational leaders are visionaries who involve bllowers to such an extent that they raise the level of human contact and ethical spirations of both leader and led. Transformational leadership looks at the basic purpose of the organisation, giving other people in the organisation a vision that will transform the xisting situation.

CHAPTER 3: RESEARCH METHODS

.1. introduction

nis chapter deals with the methods to be employed to conduct this particular research. It rill look at what the aims of the research are; the actual research questions, the specific nethods used to conduct the research and how the data is analysed.

2. The Aims of the Research

nis research project looks primarily at the extent to which transformational leadership is racticed in a Durban Secondary School. Seeing that transformational leadership is linked ith school effectiveness, (Coleman, 1994) focus of this research project is to evaluate the rincipal's leadership style. In contrast to transactional leadership, transformational adership refers to the process whereby an individual engages with others and creates a connection that raises the level of motivation and morality in both the leader and follower. The transformational leader is attentive to the needs and motives of his or her followers, ring to help them realise their full potential. Does the way the school is managed contain by evidence of transformational leadership? Is the principal's perception of his leadership yle parallel with the staff's perception of how he leads the school?

3. Research Questions

What is understood by the term Transformational Leadership?

What is its significance in the context of South African schools?

What is the principal's conception of his leadership style and to what extent does is style parallel with Transformational Leadership? How do the educators perceive the principal's leadership style? Is there a ongruence between the two perceptions?

- Based upon these perceptions, to what extent is the leadership at this school ansformational?
- . What are the implications of the findings for possible leadership development in the chool?

.4. Research Methods

conducting this research, the methods of collecting data would be both ualitative and quantitative, namely a semi-structured interview schedule and a uestionnaire. In an attempt to address the key questions, the following methods will e used:

terature regarding transformational leadership.

Luestion 3 requires a semi-structured interview with the school principal while uestion 4 will be answered by means of an interview with one deputy principal, one 10D and one level one educator. A questionnaire will be distributed to the smaining educators on the staff.

ne answer to questions 5 and 6 will be addressed by referring to questions 3 and 4 after the collected data is analysed and interpreted.

.5. Research Population

nis research project involved the entire population of educators at a Durban

econdary school, inclusive of school management. The researcher is a member of aff, but will not be part of the actual research, so the number of actual participants rould be limited to 35.

.6. The Research Instruments

ne entire research is made up of a semi-structured interview with the school principal ne deputy principal, one HOD and one level one educator. Apart from the rincipal, the other interviewees were randomly chosen, ie. 1 deputy principal from 1 two on staff, one head of department from the five and 1 post-level one ducator from 25. A questionnaire will be distributed to the remaining educators on 1 te staff.

ne Interview

ne semi-structured interview was chosen as one of the research instruments because gives each participant an opportunity to define the situation as they see it. The terview questions are structured in such a way that the respondents are able to late their opinions on leadership while giving the researcher an opportunity to see hat the respondent is thinking without the restrictions of the questionnaire.

ne principal's interview consists of 9 questions. The first question deals with how he ould define the term leadership. The aim is to bring his train of thought to the object, while letting him provide the researcher with a subjective definition of the object.

tuestion 2 deals with the principal's view of leadership and management. Does he took at the two terms as being synonymous? Are they the sole responsibility of the rincipal?

Question 3,4 and 5 question the principal on his perceptions of his abilities as a sader. How does he see himself as a leader? Where is he strong or weak? Question 6 questions the principal on whether other people see him as a leader. Question 7 questions what has influenced the principal's approach to leadership. Question 8 questions the principal's relationship with his staff while the final questions sks what he would do differently if he had the opportunity to start all over again.

ne deputy-principal, head of department and educator will be asked the same uestions. The purpose of the interview is to see the correlation between the rincipal's answers regarding his leadership style and the opinions of the staff nembers from various levels of the organisational hierarchy. The six questions in this shedule are therefore very much the same posed to the principal.

ne Questionnaire

questionnaire will be distributed to the rest of the school staff complement. The uestionnaire's primarily aimed at providing the researcher with a general staff appression of the leadership style of the school principal. Is there any evidence of ansformational leadership at the school, according to the staff?

ne questionnaire is divided into two sections the first of which concentrates on the

espondent's background information. The first 2 questions ask the number of years he respondent has been teaching at this school and altogether. The 3rd enquires as the number of institutions taught at previously while the 4th asks the present position frespondent at the school at the time of the research project. This information is nportant as it would put into perspective the past experiences of the educator, their bility to make comparisons of various leadership styles and to ascertain the position of the respondent on the organisation's hierarchy.

ne second section asks specific questions as to the opinion of the respondent agarding their principal's leadership style. The first question asks how the respondent ated their principal as a leader. Respondents are required to tick an appropriate lock ranging from numbers 1-5, 1 being excellent, 2 is good, 3 – fair, 4 – poor and 5 reing leadership is non-existent.

question 2 uses the same scale to rate the leadership of other management figures it the school. This question is aimed to see if respondents felt a certain way about the rincipal's leadership or the entire school management team.

Question 3 and 4 are open-ended questions requiring the respondents to elaborate in what they felt good leadership was. The aim here is to see if the educators know what to look for in a leader. Are there any qualities that are common? The fifth question aims to ascertain what the staff felt about the principal's elationship with his management team, staff, the parent community and the earners. Once again the same scale of 1 – 5 ranging from excellent to non-existent

. . .

ill be used to gauge the views of the respondents.

Question 6 questions the respondent if they were aware of the presence of a mission attement, vision for the school, staff policy, management plan and effective overning body. All these are crucial elements of a learning organisation and haracteristic of a school led by a transformational leader.

now 1-excellent, 2-good, 3-fair to 4-poor. Question 7.1 to 7.8 asks about the rincipal's ability to motivate, be seen, live the vision of the school, initiate teamwork, cknowledge work done well, ability to listen, value opinions and promote mutual ust. The final question allows the respondent to comment freely on leadership at neir school. Any concerns or comments would be entertained here.

7. Data Analysis

ata was captured from two sources, namely an interview and questionnaire. The terview, a qualitative method of research, will be analysed by looking at what was aid by each respondent. However, if more than one respondent made the same amment, then the frequency of the comment would then hold more water. Analysis the data from the questionnaire will primarily be via a frequency count. The openaded questions will serve to reinforce the quantitative data captured from the pestionnaire.

8. Conclusion

is chapter looked at the research methods to be employed in conducting this search project, analysing the questions used in the interview schedule and the

uestionnaire. An explanation was given as to why a question was asked. The next hapter will analyse the results of research that was conducted.

CHAPTER 4: THE FINDINGS OF THE RESEARCH

.1 INTRODUCTION

ne fourth chapter looks at the findings of the research, namely the interviews and the uestionnaires. The findings of the questionnaires will first be looked at before the interview

.2. THE QUESTIONNAIRE

ne questions were asked in a specific format and these will be analysed in that same ormat. Questionnaires were distributed to one Deputy-Principal, 4 Heads-of Department and 24 educators. All 29 questionnaires were returned.

ection A: Background Information

. How many years have you been teaching?

ne respondents were required to indicate the number of years they have been ducators. The results have been analysed as follows:

1 – 5 yrs	6 – 10 yrs	11 – 15 yrs	16 – 20 yrs	> 20 yrs	
3	3 5		4	8	

ne results show a fairly experienced staff, 72% of whom have been educators for more nan ten years.

. How many years have you been at this school? ...

ne respondents were required to indicate the number of years they have been educators t this particular institution. The results have been analysed as follows:

1 – 5 yrs	6 – 10 yrs	11 – 15 yrs	16 – 20 yrs	> 20 yrs
11	4	7	3	4

ne staffing situation is fairly stable, 62% of them being at this school for more than five ears.

. How many other institutions have you taught at previously?

ne results are as follows:

lumber of Institutions	nil	1	2	3	>3
lumber	8	7	6	5	3

lot many of the educators have taught at other institutions, but there are enough nembers who have experienced other leadership styles, having something to make a omparison with.

. What position do you currently hold at this school? (Please tick the appropriate box)

Principal	nil
Deputy Principal	1
Head of Department	4
evel 1 Educator	24

ECTION B

.How would you rate your school principal as a leader?

1-excellent 2-good 3-fair 4-poor 5-non-existent)

1	Nii	2	Five	3	Fifteen	4	Nine	5	Mil	
•	1 111		1146	5	11116611	"	MILIC	J	1411	
]						

. How would you rate the leadership of other school management figures at this school?

_					_	. —				
1	Three	2	Twelve	3	Nine	4	Four	5	One	
	111100	_	1440110		MILIC	"	1001	J	Onc	

.How would you rate the leadership of the heads of department at this school?

1 Four 2 Fourteen	3 Nine	4 One	5 One
-------------------	--------	-------	-------

ne majority of the staff felt that the school management team (SMT) were better leaders nan the principal himself. 31% of the respondents said that the principal was a poor eader, while more than 50% felt that the SMT were good or excellent.

. What would you regard as good leadership? (Comment briefly)

nis question allowed the respondents to comment on what they felt good leadership was ne respondents were not asked to merely tick off from a list of qualities, but rather think of ualities of a good leader. More than one respondent listed the following qualities:

Leadership Quality	No. of Respondents
Leads by example.	7
Listens and motivates	6
Has a hands-on approach	5
Able to delegate	5
Having a vision	5
Consults before making decisions	4
Is visible around the school	4
Have the ability to make others be the best	
that they can be	3
Is a person that is approachable	3
Is an effective communicator	2
A collaborative, participative decision-maker	2
Not afraid of change	2
Able to communicate at all levels 34	2

other comments relating to good leadership were, Balanced individual, A person aware f his/her strengths and weaknesses; one mature enough to accept advice/input from olleagues without feeling threatened; a person who is transparent; acknowledges work rell done; is an independent thinker and is a support structure. These responses show that he educators are aware of what qualities to look for in a leader.

. How would you rate your principal's relationship with the following?

-excellent 2-good 3-fair 4-poor 5-non-existent)

1. School Management

1	One	2	Twelve	3	Ten	4	Five	5	One

2. Educators

Г	_			<u></u>	T		T .	•	1 -	
	J	One	2	Seven	3	Inineen	4	SIX	5	Iwo

3. Parents

1	Three	2	Ten	3	Fourteen	4	Two	5	Nil	
•	111100		1011	•	100110011	7	1440	7	140	

4. Learners

1	Two	2	Eight	3	Fifteen	4	Four	5	Nil	

ne results of question 5 indicate that the principal has a good relationship with all roleayers at the school. Seven of the respondents felt that he had an excellent with one or ore of the listed stakeholders.

To the best of your knowledge, does your school have?

Yes	No	Total
9	20	29
12	17	29
2	27	29
6	23	29
11	18	29
_	9 12 2 6	9 20 12 17 2 27 6 23

ven though the principal's interpersonal relationships were very good, good leadership is nown when the principal has the above in place. Sterling et al (2000) assigns functions like rategic planning, building of the school's vision, liaison with communities and parents and development of the staff can be identified as the functions of a leader. Schools are bliged to present a School Management Plan, setting out priorities for development and ow they go about implementing them as well as anticipating future resources. However, reating a plan is not the same as creating a vision [Jenkins, 1991). Planning has to do with the problems of today and tomorrow, prioritising and allocating resources on a short-term asis. The results show that the majority of the staff is not aware of these things happening I this school. Maybe none of them are in place or maybe they are, but many are not ware of them. Remember, however, vision cannot be imposed. Bennis and Nanus (1985) arn that a vision can only be created successfully if it grows out of the needs of the entire ganisation and is owned by all the important stakeholders.

How would you rate your principal with regards to the following?

|-excellent 2-good 3-fair 4-poor)

	1	2	3	4	Total
.1. Motivating the staff to give of their best	0	5	9	15	29
.2. Highly visible throughout the day	0	2	4	23	29
.3. Constantly living the vision of the school	0	3	11	13	29
.4. Getting staff to work together as a team	0	3	9	16	29
.5. Acknowledges work well done	0	6	11	12	29
.6. Listens carefully to the needs of his followers	0	7	10	12	29
.7. Values the opinion of others when making					
decisions.	0	3	5	21	29
.8. Promotes mutual trust between staff and		***			
management	0	3	13	13	29

he general ratings ranged from fair to poor with not one respondent rating the principal is excellent in the above qualities. These are the characteristics of good leadership as lighlighted by the accompanying literature, so in evaluating the principal as a leader, aducators do not see him as being one.

i. Is there any comment that you would like to make regarding leadership at this school? Here the respondents were required to make any comment they felt was pertinent to the context of the study in question. This question allowed them to make comments on eadership at this school about which they felt strongly and was not asked to in the questionnaire. The most popular answers are listed below:

Comment	No. of Respondents

ne principal at this school must be visible.	5	-
eadership at this school is non-existent	3	
ne principal's leadership style is autocratic.	3	
nere is a breakdown in communication between		
nanagement and the rest of the staff.	3	
art of the school management team do a good job	2	<u>. </u>

Other individual comments are now listed,

- The principal has a good one-to-one relationship with the staff, but fails to address the staff as a whole.
- The staff of this school also needs to work with the school management, being professional in going about their job.
- Educators need to show respect/ be less confrontational.
- We never have staff meetings.
- The principal of this school is trustworthy, mannerly and has tact. He is, however, paranoid regarding instructions from the Department of Education; being in fear of doing the wrong thing.

1.3. Summary

Looking at the responses to the questionnaire, the staff saw the SMT as better leaders than the principal. These educators have shown a good understanding of what good eadership is and even though the principal had a very good relationship with the SMT,

ducators, pupils and parents, they rated him fair to poor when it came to enacting the lements of transformational leadership.

.4. The Interviews

In interview was conducted with the principal, one of the deputy-principals, a head-ofepartment and a level one educator. Analysis of the interviews follow:

RINCIPAL

.How would you define the term leadership?

ne principal responded that a leader is the head of an organisation, saying that there vere 3 aspects to the concept namely, management and administration, human relations and vision.

. Is leadership and management the sole responsibility of the principal? Explain.

to. The organisational objective is to be a high quality institute of learning. This is only schieved through the joint effort of the School Management Team, governing body and he educators. Here the principal admitted to the role of other role-players in managing he organisation.

3. What would you describe as your strengths and weaknesses?

he principal felt that his strengths included capacity for work, his ability to multitask, project management, his commitment to community building and transformation. He has a good professional relationship with parents, teachers, the education department and he community.

te felt that his weaknesses were his high expectations of others and the fact that he did not delegate enough.

ne principal also felt that he had insufficient contact with the educators, admitting that e needed greater involvement in staff development.

. What would you regard as the critical points in your career?

979 – as a locum, taking a conscious decision to make teaching my career.

981 – Top school (House of Representatives) for Physical Science (Higher & Standard Frade)

982 – British Council Award to attend the British Association of Science Educators Conference in the United Kingdom.

984 – Appointment as a Head of Department (Science)

. If you had to describe your leadership style, what would it be?

ong-term development. Looking at the bigger picture to develop the school over a period of time. The principal said that he had a vision as to where the school would be eventually.

. How would you say others view you regarding your leadership style?

Demanding high standards with attention to detail. He felt that others saw him as a perfectionist, wanting to do everything on his own.

1. What past events most influenced your leadership approach?

The South African Constitution and the Bill of Rights because it redefined the whole approach to education in South Africa, particularly the central role of the learner in education.

3. How would you regard your relationship with your staff?

The principal valued the staff for their commitment to the learners and the school especially in light of the difficult working conditions. The personal relationships are sound. The principal empathised with and understood the many difficulties faced in the

6 ♦ 40

lassroom and in their many personal situations. He tries to balance this with a sound rofessional relationship with all educators.

. If you, as a leader, had to do something different, what would it be?

ne principal felt that he would prefer to be a leader in the classroom situation, as possed to being the school principal. He felt that being in the classroom was the reason nat he became an educator.

n summarising the interview, it is clear that the principal understood the challenges facing outh African schools today, knowing what is expected of him as principal. He admitted nat to be an effective learning organisation, then all role-players needed to be involved n managing the organisation. He also seemed to understand what qualities a good eader possessed. He admitted that he did not delegate enough. A good school leader vould therefore move away from the token attempts at staff involvement to more radical orms of power sharing. A way to do this would be to create highly autonomous teams to whom power is granted on a self-managing basis.

te had a vision, but did not communicate it to everyone. Vision is an interpretation of the direction in which an organisation is going, something necessary to give people a belief in the organisation's future. It is therefore senseless having a vision if none of the stakeholders are aware of this vision.

te reinforced what the educators said about his interpersonal skills, but felt that his place would rather be in the classroom as opposed to the principal's office.

<u>DEPUTY-PRINCIPAL</u>, HEAD OF DEPARTMENT (HOD) AND EDUCATOR'S INTERVIEW

1. How would you define the term leadership?

The deputy principal said that a leader is defined as the person at the helm, leading by example and delegating in a meaningful way for the benefit of the whole institution. The

OD's explanation of a leader was that he/she is a motivator while the educator's efinition of the term included good interpersonal skills, initiative, able to handle conflict nd be impartial.

. What would you regard as good leadership?

lere the three respondents said that a good leader had good interpersonal relationships, leals with conflict, is impartial, is both sympathetic and empathetic and keeps their ersonal perspective out of decision-making. One respondent noted that a good leader rould look for strengths in your staff, harnessing these strengths while taking note of any reaknesses.

Is leadership and management the sole responsibility of the principal? Explain.

If three respondents said no. There should be participation at all levels with the principal lelegating, looking for strengths that he himself does not have.

One respondent felt that there should, however, be someone in the driver's seat.

1. What would you describe as your principal's strengths and weaknesses?

he principal's strengths included being a good listener, approachable, sympathetic to earners and staff on a personal level, tolerant, compassionate and caring. The espondents felt that the weaknesses included being absolutely clinical, taking on too nuch on his own without delegating, is aloof and does not see himself as part of the staff.

i. If you had to describe your principal's leadership style, what would it be?

One respondent said that the principal had no leadership style. There was no notable evidence of leadership from him. Another felt that the principal was autocratic with very little faith in his management team.

5. How would you regard the principal's relationship with his staff?



It the respondents felt that the relationship with the staff was not a healthy one. The slationship was not conducive to teamwork and not in the best interest of the school as a rhole. One-on-one, the principal is very friendly, but in a group he seems to avoid onfrontation. One respondent intimated that the school operated because the ducators were hardworking.

ne interviews showed that the interviewees understood what to look for in a leader.

eadership and management was, they said, the task of all stakeholders with the principal it the helm. They saw the principal as a compassionate, caring person whose downfall was that he took on far too much on his own. His professional relationship with the staff wa of healthy and not conducive to teamwork. None of the respondents felt that the principal possessed any leadership qualities.

.5. Overview of the Findings

even though there were a few respondents who saw the principal as a leader, the sadership characteristics shown were more transactional than transformational. However he majority of the staff felt that the principal did not possess any of the qualities required of transformational leadership. They were not aware of any vision for the school, there is not chool Management Plan that they know of, staff don't feel empowered and many feel titled when it comes to their creativity. School leadership has to play the role of consultant, facilitator, counsellor and empowerer, ensuring that teams work effectively, while giving them the necessary back up and support in terms of time and technical esources.

'he principal did not claim to be a transformational leader. He did not even say that he was a good leader, demonstrating that he knew what good leadership was. He did,

owever, feel that he was managing the school effectively bearing in mind the lack of sources and other obstacles before him. He expected a lot from his subordinates, but as reluctant to delegate with the result that he has lost contact with his staff rofessionally, resulting in frustrations from many organisational stakeholders.

.6. Conclusion

nis chapter has concentrated on analysing the responses to the questionnaire and the interviews that were conducted. The findings of the research will be discussed in the next hapter followed by the recommendations based on the findings of the research.



CHAPTER 5: THE FINDINGS OF THE RESEARCH AND RECOMMENDATIONS

1. Introduction

nis chapter summarises the findings of the research conducted in this study, an interview with the principal, a deputy-principal, an HOD and an educator as well as a questionnaire istributed to the remainder of the staff. Coupled with this, the research questions will be unmarized before making any recommendations as to the way forward for this school.

.2. Summary of the Findings

nis research project focussed on transformational leadership. Was there any evidence of ansformational leadership at a Durban Secondary school?

ne first research question asked what was understood by the term transformational eadership, while the second questioned its significance in the context of South African chools. This question was addressed by the literature review, which highlighted the fact nat transformational leadership engages leaders and followers in such a way as to lift one mother to higher levels of motivation and morality. The transformational leader, according to Bass (1990), provides a vision and sense of mission in his/her followers, notivating them to become part of the organisation's shared vision. The significance of his kind of leadership is that it promotes a culture of teaching and learning, the primary ask of a school leader.

he school manager in South Africa is now faced with far greater responsibility than ever refore. Schools need effective and efficient schools management that are equipped with new and improved skills, knowledge and attitudes to help them cope with a range of rew demands and challenges. Managers must work more closely with parents and the

ommunity at large, assume greater financial responsibility, cope with a multicultural chool population, and manage change and conflict while making do with fewer sources. "The extent to which these schools are able to make the necessary changes epends largely on the nature and quality of their internal management." (Department of ducation, 1996: 28). It is the place of transformational leaders to take up the challenge acing South African school principals. These leaders are visionaries, masterful ommunicators with the ability to inspire trust in their followers while making them feel apable.

ne third question asked what the principal's conception of his leadership style was and to that extent did this style parallel with Transformational Leadership. A semi-structured nterview was conducted with the principal and it was discovered that he had a particula ong-term vision for the school and he led in such a way so as to achieve those goals. He acknowledged that the organisational objective is to be a high quality institute of learning his is only achieved through the joint effort of the School Management Team, governing and administration. Also, he did see leadership as being made up of management and administration, human relations and vision. The vision aspect of leadership in this case eemed to be the principal's personal vision for the school as many of the educators were not aware of it.

He saw others viewing him as a person who demanded high standards. He felt that his personal relationship with the staff was sound and he tried to balance this with a sound professional relationship.



The fourth question looked at how do the educators perceived the principal's leadership yle. Was there congruence between the two perceptions? Here a semi-structured terview was conducted with a deputy-principal, a head-of-department and a level 1 ducator. A questionnaire was administered to the remainder of the staff to gather the equired data for the research. More than 50% of the educators did not see any evidence f transformational leadership in the principal's style. Most educators felt that the principal id not motivate the staff to give of their best, was not visible during the school day, did of make people aware of the vision for the school, does not promote teamwork or value ne opinions of other educators.

is clear from the interviews and questionnaire that the principal did not claim to be ansformational in his leadership style and the staff felt that he was a bit autocratic as a sader.

Question 5 asked to what extent leadership at this school was transformational. There was the evidence of transformational leadership at this school, so it is safe to say that there is nuch to be learnt concerning leadership as a topic.. The extent to which schools will be able to make the transition to being a self-managed organisation depends a great deal pon the nature and quality of their internal management. (Department of Education 996). The absence of any transformational leadership characteristics at this school means not there would be very little improvement in their performance 'MacBeath (1998) clearly adicates that leadership has the task of ensuring that the school is learning. Leadership willds conditions for reflection, open dialogue and mutual respect for everybody's ideas. Individuals, groups, the institution as a whole, other role-players and school management would, in a learning organisation, be allowed to grow and develop.

3. Recommendations

poking at what has come out of this research project, it is clear that the principal's padership style is called into question. Some educators felt that the principal was utocratic while others saw the school as being without leadership.

is recommended that the principal ensure that the relevant parties viz. educators, schoo overnors, parents and SMT draft a new vision and mission statement using the guidelines s set out in the South African Schools Act. Beare et al (1993) also says that a pivotal activity of a leader is to constantly engage in the ever-changing process of stating a visior of things to come. The principal would constantly be asking: "Where are we and where are we going?" Thereafter ask: "How are we going to get there?"

econdly, expose all those in a leadership role at the school to the many leadership and nanagement courses available to them so that they are aware of other leadership styles and can learn valuable skills like how to delegate, managing people, managing change, and so on.

he third recommendation deals with the principal's own leadership style. The characteristics of transformational leadership and good leadership are referred to in this esearch project. An attempt could be made to incorporate some of these into his own nanagement style.

Fourthly, what has come out of the research is the fact that the principal is not always accessible. The principal must attempt to make himself accessible to his staff, adopting an appendoor policy, thus promoting a culture of collaboration.



astly, as a transformational leader the principal should constantly be motivating his staff improve themselves and by so doing, improve the organisation. He would be there to rovide support whenever called upon for it.

.4. Conclusion

ne literature reviewed in this project has highlighted the challenges facing South African chools today. These newfound challenges, research has shown, are easier to overcome if chool management are transformational in their leadership style. Transformational leaders are a vision for the organisation, ensuring that all role-players are aware of that vision and systematically works towards attaining those goals. This type of leadership is directly elated to school effectiveness and school improvement.

ne findings of this research have shown that there is no evidence of transformational adership at this school. The recommendations have highlighted ways in which the trincipal can begin to transform the school so that they too may improve to such an extent that the school is seen as being effective.



REFERENCES

ass, B. & Avolio, B. (1994) Improving Organisational Effectiveness Through Transformational advership, London: Sage.

ass, B.M. (1990), Bass and Stogdill's handbook of Leadership: theory, research and nanagerial applications, New York: Free Press.

ennett, N., Crawford, M. and Riches, C. (1993) "Introduction: Managing Educational hange: The Centrality of Values and Meanings" (pp4-8) in Preedy, M. (ed), *Managing***Reference School, London: Paul Chapman Publishing Ltd.

eare, H. Caldwell, B.J. & Millikan, R.H. (1989) **Creating an Excellent School**, London: outledge.

ennis, W. & Nanus, B. (1985), **Leaders: the strategies for taking charge**, New York: Harper & ow.

ryman, A. (1992) Charisma & Leadership in Organisations, Sage Publications: London.

ush , T. (1994). "Theory and Practice in Educational Management" in Bush, T. & Westurnham, J.(eds) *The Principles of Educational Management*, Harlow: Longman.

ush T. Coleman M. & Thurlow M. (eds) (2003) Leadership and Strategic Management in outh African Schools, London: Commonwealth Secretariat.

aldwell, B. & Spinks, J.(1992) Leading Self-Managing Schools, London: Falmer Press.

carnall, C.A. (2003) Managing Change in Organisations 4th ed., Harlow: Prentice Hall.

cave, E and Wilkonson, C. (1993) "Developing Managerial Capabilities in Education" in reedy, M. (ed), *Managing the Effective School*, London: Paul Chapman Publishing Ltd.

Freemers, B.P.M. (1994) Effective Classroom, London: Cassell.

Pay, C; Harris, A; Hadfield, M; Tolley H & Beresford J. (2000) Leading Schools in Times of Change. Trowbridge: The Cromwell Press

Oubrin, A.J. (2001) Leadership: Research Findings, Practice, and Skills, Boston: Houghton Aifflin Company.

Dunford, J; Fawcett R; Bennett, D. (2000) School leadership: National And International Perspectives. Great Britain: Clays Ltd

Frace, G. [1995] **School Leadership: Beyond Education Management. An Essay in Policy** icholarship, London: The Falmer Press.

lenkins, H.O. (1991), **Getting it Right: A Handbook for Successful School Leadership**, Comwall: T.J. Press Ltd.

NacBeath, J. (ed.) (1998) Effective School Leadership: Responding to Change, London: aul Chapman Publishing.

lorthouse, P.G. (2001) Leadership: Theory and Practice 2nd ed., Thousand Oaks: Sage sublications.

wens R.G. (1998) Organizational Behaviour in Education. (6th ed.) Boston: Allyn and acon.

he Report of the Task Team on Education Management Development: **Changing tanagement to Manage Change in Education** (1996), Pretoria: Department of Education.

chon, D.A. (1983), The Reflective Practitioner: how professionals think in action, London: emple Smith.

ergiovanni ,T. & Starratt, R.J. (1988), **Supervision: Human Perspectives**, New York: Mc Graw fill.

iterling L & Davidoff S.(2000). The Courage to Lead: A whole School Development Approach. Kenwyn: Juta & Co Ltd.

iquelch, Lemmer (1994) Eight Keys to Effective School Management in South Africa, iouthern Book Publishers: Halfway House.

/est-Bumham, J. (1994), "Management in Educational Organisations" in Bush, T. & Westurnham, J.(eds) *The Principles of Educational Management*, Harlow: Longman.



DDENDUM 1

<u>UESTIONNAIRE</u>: Transformational Leadership

am a M.Ed. student at the University of Natal and I have chosen to research the above spic. Part of my research involves gathering data from you as staff members, so I would really appreciate your cooperation in completing this questionnaire.

<u>lease note:</u>

- Any information divulged in this questionnaire is confidential. Please do not write your name anywhere on this questionnaire.
- 2. As far as possible, please answer all the questions as accurately and as honestly as possible.
- 3. I once again thank you in anticipation for your cooperation.

section A: Background Information

. How many years have you been teachi	ng?
2. How many years have you been at this	school?
3. How many other institutions have you to	aught at previously?
\$. What position do you currently hold at t	his school?(Please tick the appropriate box)
Principal	
Deputy Principal	
Head of Department	
Level 1 Educator	

<u>section B:</u>

This is not a test of competence. Please mark your opinion with the appropriate number on

ne scale provided fo	or each question.			
o what extent do y	ou agree/disagree (with questions 1 - 2	2?	
.How would you ra	ite your school prir	ncipal as a leader	ş	
(1-excellent 2-go	ood 3-fair 4-pc	oor 5-non-exister	nt)	
1	2	3	4	5
. How would you ro	ate the leadership	of the deputy-prir	ncipals at this scho	O \$
1	2	3	4	5
i.How would you ra				
1	2	3	4	5
1 M/h at would your		rdorchin? Ioommo	ent brinsh A	
I.What would you re	egara as good i e c	ideisnips (comme	en onenyj	
	••••••			***************************************
5. How would you re	ate your principal's	s relationship with	the following?	
1-excellent 2-god	od 3-fair 4-pod	or 5-non-existen	†)	
5.1. School Manage	ement			
1	2	3	4	5
5.2. Educators				
1	2	3	4	5
	1			



.3. Parents

ì	2	3	4	5	
.4. Learners					
,4. Leomers					
1	2	3	4	5	

.. To the best of your knowledge, does your school have?

	Yes	No
A mission statement		
A vision		
A policy on Staff Development		
A School Management Plan		
An effective School Governing		
Body		

7. How would you rate your principal with regards to the following?

1-excellent 2-good 3-fair 4-poor)

7.1. Motivating the staff to give of their best	1	2	3	4
7.2. Highly visible throughout the day	l	2	3	4
7.3. Constantly living the vision of the school	1	2	3	4
7.4. Getting staff to work fogether as a team	1	2	3	4
7.5. Acknowledges work well done	1	2	3	4
7.6. Listens carefully to the needs of his followers	1	2	3	4

./. Values the opinion of others when making	
decisions.	1 2 3 4
.8. Promotes mutual trust between staff and management	1 2 3 4
8. Is there any comment that you would like to make regardschool?	rding leadership at this



DDENDUM 2

ITERVIEW SCHEDULE (PRINCIPAL)

.How would you define the term leadership?
. Is leadership and management the sole responsibility of the principal? Explain
. What would you describe as your strengths and weaknesses?
. What would you regard as the critical points in your career?
i. If you had to describe your leadership style, what would it be?
i. How would you say others view you regarding your leadership style?
7. What past events most influenced your leadership approach?
3. How would you regard your relationship with your staff?
). If you, as a leader, had to do something different, what would it be?

DDENDUM 3

NTERVIEW SCHEDULE (DEPUTY-PRINCIPAL, HEAD OF DEPARTMENT AND 1 EDUCATOR)

- .How would you define the term leadership?
- :. What would you regard as good leadership?
- i. Is leadership and management the sole responsibility of the principal? Explain.
- I. What would you describe as your principal's strengths and weaknesses?
- i. If you had to describe your principal's leadership style, what would it be?
- 5. How would you regard the principal's relationship with his staff?

